Dear editors:
On behalf of my co-authors, thank you very much for giving us an opportunity to revise our manuscript, we appreciate reviewers and editors very much for your great efforts on our manuscript.
For formatting issues, we have made changes one by one as required. We adjusted the table to a three-line table and inserted references after the names of the authors in the table. Meanwhile, we reprocessed the figures and uploaded the files as required.
For language problems in the manuscript, we have re-edited the revised manuscript by a professional English language editing company and have provided a new language certificate along with the manuscript.
We would like to express our great appreciation to you on our paper. Looking forward to hearing from you.

Yours sincerely,
Baoding Chen

Reviewer Comments:
Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: This is an interesting case-report, that gives potentially useful suggestion to palliate neoplastic masses of the eyelid. I have no criticism about it, but I would suggest to temper the conclusion of the abstract: it is just a case report, and further experiences should be added before proposing MWA as a safe and effective method to treat eyelid masses (as correctly stated by the authors in the Conclusion of the full text).

Response:
Thank you very much for taking your time to review this manuscript.
Thanks for your suggestion on the part of abstract. Rechecking the manuscript, we found that the conclusion section of the abstract was inappropriately stated. We have revised this part to only state the following two points:
1. Ultrasound-guided microwave ablation is an optional method for the management of mycosis fungoides eyelid involvement.
2. Ultrasound-guided microwave ablation is effective for the patient in this case report.
Furthermore, we have polished the language in the manuscript and corrected some language problems.
We appreciate for your warm work earnestly again, and hope that the correction will meet with approval.